\documentclass[11pt]{handout}
\usepackage{6044}
%\documentclass[11pt]{article}
%\usepackage{6001}
\usepackage{time,fullpage}
\usepackage{amsmath}

\begin{document}
\handout{ps1}{Problem Set 1}{9 September 1996}

\due 16 September 1995.

\medskip

{\bf Reading for this problem set}: Handout 3, ``Notes on Proving
Arithmetic Equations.''

{\bf Reading for next week:} Handout 2, ``A Term Rewriting Model for Scheme, I.''
\medskip

Do the following problems from Handout 3:

{\em Handout 3, Problems}: 3, 4, 5, 6(c), 6(f), 10, 11.

For each problem or problem part, you may assume all the results in the
preceding notes or problems.  Do not be discouraged by the discussion
question, Problem 11.  Write a brief response in preparation for a class
discussion on Monday, Sept. 16.

\end{document}
